Author's response to reviews

Title: Contraceptive use and associated factors in Afghanistan: A secondary analysis of Afghanistan Mortality Survey (AMS) data

Authors:

Mohammad H Rasooly (dochafez@yahoo.com)
Mohammad Ali (alim@who.int)
Nick JW Brown (nickiwbrown@gmail.com)
Bashir Noormal (dgaphi.moph@gmail.com)

Version: 2 Date: 16 June 2014

Author's response to reviews: see over
Dear Editor in Chief,

Please find enclosed our revised manuscript “Contraceptive use and associated factors in Afghanistan: A secondary analysis of Afghanistan Mortality Survey (AMS) data” by Mohammad Hafiz Rasooly et al which we would like to submit for publication as a original research article in BMC Women’s Health”. To our knowledge this is the first report showing the prevalence and associated factors of contraceptive use in Afghanistan. In addition, this manuscript describes the contraceptive use by provinces and regions in the country. We believe our findings would appeal to the leadership of BMC Women’s Health.

We would like to confirm that our manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the manuscript and agree with its submission to BMC Women’s Health.

Please address all correspondence to dochafez@yahoo.com. We look forward to hearing from you at your earliest convenience.

Sincerely,

Mohammad Hafiz Rasooly, MD, MSc
Correspondence author /Technical Advisor
Afghanistan National Public Health Institute
Ministry of Public Health Afghanistan